Author's response to reviews

Title: Early impact of an integrated maternal, newborn, and child health program, Northern Nigeria, 2009 to 2011

Authors:

Sally E Findley (sef5@columbia.edu)
Omolara T Uwemedimo (oot2101@columbia.edu)
Henry V Doctor (hvd2105@columbia.edu)
Cathy Green (cgreen@healthpartners-int.co.uk)
Fatima Adamu (fladamuy@yahoo.com)
Godwin Afenyadu (gafenyadu@gmail.com)

Version: 2 Date: 4 April 2013

Author's response to reviews: see over
April 4, 2013
Dear Ms. Reyes,

As you have requested, we have added the following sentences regarding the ethical review process used for approving the study.

The study was approved by the Katsina State Ethics Review Committee, the Yobe State Ethics Review Committee and the Zamfara State Ethics Review Committee, as both a cross-state and individual state approval. These ethics review committees are certified by the Nigerian Federal Government's National Health Research Ethics Committee to review and approve health research protocols for their states.

We look forward to hearing from the reviewers, as we think it a very important study.

Sincerely yours,

Sally E. Findley
Professor of Clinical Population and Family Health and of Clinical Socio-Medical Sciences
Mailman School of Public Health
Columbia University